OPTION AGREEMENT

BETWEEN:

ALEX PLESON

– and –

1257590 B.C. LTD.

Concerning:

Mining Claims – Northwestern Ontario, Canada

Table of Contents

<u>Page</u>

ARTICLE 1 1.1 1.2	DEFINITIONS AND PRINCIPLES OF INTERPRETATION Definitions Schedules	1
ARTICLE 2	REPRESENTATIONS AND WARRANTIES	.10
2.1 2.2 2.3	Vendor Representations and Warranties Mutual Representations and Warranties Survival and Investigation	.12
ARTICLE 3	OPTION	.14
3.1 3.2 3.3 3.4 3.5 3.6	Grant of Option No Obligation Cash in Lieu and Excess Expenditures Accelerated Expenditures Share Issuance Matters Notice of Option	14 16 16 17 17
ARTICLE 4	VESTING OF INTEREST AND SUBSEQUENT MATTERS	.17
4.1 4.2	Vesting of 100% Interest Royalty	
ARTICLE 5	OPERATOR AND OPERATIONS	.18
5.1 5.2 5.3 5.4 5.5 5.6	Operator General Duties of Operator Removal of Operator Work Programs and Budgets Emergency or Unexpected Funding Maintenance Payments	.18 .20 .20 .20
ARTICLE 6	AREA OF INTEREST	.21
6.1 6.2 6.3 6.4 6.5	Area of Interest Prohibition on Acquisitions Notice and Election. Costs and Expenses No Enlargement of Area of Interest	.21 .21 .22
ARTICLE 7	RESTRICTIONS ON TRANSFER	.22
7.1	Restriction on Transfer	.22
ARTICLE 8	TERMINATION AND SURVIVAL	
8.1 8.2	Termination Survival	
ARTICLE 9	EVENT OF FORCE MAJEURE	
9.1	Event of Force Majeure	.24
ARTICLE 10	CONFIDENTIALITY	
10.1 10.2	Confidentiality of Information Permitted Disclosure	

10.3	Exception	25
ARTICLE 11	GENERAL	25
11.1	Rules of Interpretation	
11.2	Arbitration	
11.3	Entire Agreement	
11.4	Parties' Rights to Conduct Other Business	
11.5	No Reliance or Inducement	
11.6	Tax Included and Refundable Tax Credits	
11.7	Applicable Law	
11.8	Expenses	
11.9	Notices	
11.10	Assignment and Successors	
11.11	Severability	
11.12	Waiver and Amendment	
11.13	Relationship of the Parties	
11.14	Further Assurances	
11.15	1Execution in Counterparts and by Electronic Delivery	33
Sched	ule A	A-1
Sched	ule B	B-1

OPTION AGREEMENT

THIS AGREEMENT, made effective as of the <u>5th</u> day of <u>August</u>, 2020

BETWEEN:

ALEX PLESON, an individual resident in the Province of Ontario,

(the "Vendor")

- and –

1257590 B.C. LTD., a corporation incorporated under the laws of the Province of British Columbia,

(the "Purchaser")

(collectively, the "Parties" and each, a "Party")

WITNESSETH THAT:

WHEREAS the Vendor is the registered and beneficial holder of 100% of the right, title and interest in and to the Property (as defined herein);

AND WHEREAS the Parties wish to enter into this Agreement in order to provide for the grant to the Purchaser of the Option (as defined herein) to acquire from the Vendor a 100% interest in the Property (subject to the Royalty (as defined herein)) all for the consideration and upon the terms and conditions set forth herein;

NOW THEREFORE in consideration of the mutual covenants herein contained, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties agree as follows:

ARTICLE 1 DEFINITIONS AND PRINCIPLES OF INTERPRETATION

1.1 <u>Definitions</u>

Capitalized words and phrases used in this Agreement shall have the meaning given to such words and phrases below:

"**Aboriginal Peoples**" means any Indian, first nation, Inuit, Métis or indigenous persons or groups, or first nations, bands or any other such similar legal entities.

"Acquisition Costs" has the meaning set out in Section 6.3.

"Acquisition Date" has the meaning set out in Section 3.1.

"Additional Mineral Rights" has the meaning set out in Section 6.2(a).

"Additional Other Rights" has the meaning set out in Section 6.2(b).

"Affiliate" has, with respect to the relationship between two or more bodies corporate, the meaning given to it in the *Business Corporations Act* (British Columbia) as such act may be amended from time to time and, with respect to the relationship between two or more Persons any of which are not bodies corporate, a Person shall be deemed to be an Affiliate of another Person if one of them is controlled by the other or if both are controlled by the same Person, and for this purpose, control means the right, directly or indirectly, to direct or cause the direction of the management of the affairs of a Person, whether by ownership of securities, by contract or otherwise.

"**Agreement**" means this Option Agreement, including all schedules, and all instruments supplementing, amending or confirming this Agreement and references to "Article" or "Section" are to the specified article or section of this Agreement.

"Applicable Law" or "Applicable Laws" means any applicable federal, provincial, or local statute, regulation, rule, by-law, ordinance, order, policy or consent, including the common law, as well as any other enactment, treaty, official directive or guideline issued by a Governmental Authority (including, without limitation, Environmental Laws, the Securities Act, and the policies of the Exchange) and the terms and conditions of any permit, licence or similar document or approval issued by a Governmental Authority, and shall also include any order, judgment, decree, injunction, ruling, award or declaration, or other decision of whatsoever nature of a court, administrative or quasi-judicial tribunal, an arbitrator or arbitration panel or a Governmental Authority of competent jurisdiction that is not subject to appeal or that has not been appealed within the requisite time therefor.

"Area of Interest" has the meaning set out in Section 6.1.

"**Business Day**" means any day other than a Saturday, Sunday or day that is a statutory holiday in any of Vancouver, British Columbia or Toronto, Ontario.

"Cash Payments" has the meaning set out in Section 3.1(a).

"Claim" means any claim, demand, action, cause of action, damage, loss, cost, liability or expense, including reasonable Costs incurred in investigating or pursuing any of the foregoing or any proceeding relating to any of the foregoing.

"**Common Shares**" means the common shares in the capital of the Purchaser (or resulting issuer, in connection with an IPO).

"Confidential Information" has the meaning set out in Section 10.1.

"**Contract**" means any contract, agreement, instrument or commitment, whether oral or written, that relates to the Property, including, without limitation, those to which the Vendor is bound or in respect of which the Vendor may have liability.

"**Costs**" means any and all damages, claims, losses, including economic losses, costs, expenses, liabilities and obligations of whatsoever kind, direct or indirect, including fines, penalties, interest, lawyers' fees and disbursements, expenses and taxes thereon.

"Designee" has the meaning set out in Section 11.2(e).

"Effective Date" means the date of this Agreement, as first above written.

"Encumbrances" means any lien or deemed trust (statutory or otherwise), charge, hypothecation, pledge, mortgage, title retention agreement or arrangement, conditional sale agreement, right of set-off or arrangement, option or earn-in agreement, covenant, condition, lease, license, security interest of any nature, claim, exception, reservation, easement, encroachment, right of occupation, right-of-way, right-of-entry, matter capable of registration against title, assignment, right of pre-emption, royalty, right, privilege or any other encumbrance or other adverse third party interest of any nature (including any execution, seizure, attachment or garnishment which binds property), regardless of form, whether or not registered or registrable and whether or not consensual or arising by any Applicable Law, and includes an agreement to give or create any of the foregoing.

"Environmental Laws" means all Applicable Laws relating to the protection of the environment, including air, soil, surface water, ground water, land, biota, wildlife or personal or real property, or to employee and public health and safety, and includes those Environmental Laws that regulate, ascribe, provide for or pertain to liabilities or obligations in relation to the existence, use, production, manufacture, processing, distribution, transport, handling, storage, removal, treatment, disposal, emission, discharge, migration, seepage, leakage, spillage or release of Substances, pollution, noise or the construction, alteration, use or operation, closure, dismantling, abandonment, demolition or decommissioning of any facilities, mines, workings or other real or personal property and the reclamation or restoration of lands.

"**Event of Force Majeure**" means any event or circumstance, or a combination of events and/or circumstances:

- (a) that causes or results in the prevention or delay of a Party from performing any of its obligations in this Agreement;
- (b) which is beyond the reasonable control of that Party; and
- (c) could not, or the effects of that event or circumstance could not, have been prevented or delayed, overcome or remedied by the relevant party acting reasonably,

and, provided the event or circumstance meets the foregoing criteria, includes:

- acts of war (whether war be declared or not); public disorders, insurrection, rebellion, revolution, terrorist acts, sabotage, riots or violent demonstrations;
- civil disobedience, caused by indigenous peoples, environmental lobbyists, non-governmental organizations or local community groups or other Persons;
- (iii) injunctions imposed by any governmental authority except if caused by a breach of Applicable Law or a court resolution;
- (iv) explosions, fires or floods not caused by or attributable to a Party;

- (v) floods, earthquakes, hurricanes or other natural calamities or acts of God;
- (vi) shortages in workforce or supplies, travel and access restrictions imposed by Governmental Authority or other Third Parties, or other delays caused by endemics, epidemics or pandemics;
- (vii) strike or lockout or other industrial labour action or disruption (including unlawful, but excluding lawful strikes or lockouts or other industrial labour action) which:
 - (A) has national, regional, provincial or state-wide application,
 - (B) directly affects the performance of the obligations under this Agreement, and
 - (C) lasts for more than seven consecutive calendar days;
- (viii) any action or failure to act within a reasonable time without justifiable cause by any Governmental Authority, its employees or agents including the denial of or delay in granting any land tenure, concession, authorization, licence, permit, lease, consent, approval or right which denial or delay shall imply a material adverse effect on the construction or operation of any project in respect of the Property, upon due application and diligent effort by the Party to obtain same, or the failure once granted to remain in full force and effect or to be renewed on substantially similar terms;
- (ix) denial of access to the Property by any surface-landowner in the area where the Property is located;
- (x) interference with Property by way of delaying the issuance of any land tenure, concession, authorization, licence, permit, lease, consent, approval or right or by denying access to the Property, by Aboriginal Peoples in connection with land claims or other disputes relating thereto, or other occupants of the lands comprising, or around, the Property; and
- (xi) injunctions (granted by a court or other Governmental Authority) not caused by any breach of this Agreement by any Party whether of the kind enumerated above or whether foreseen, foreseeable or otherwise unforeseeable.

"Exchange" means a recognized stock exchange in Canada.

"**Expenditures**" means all costs and expenses funded, spent or incurred in the conduct of activities directly on, or in relation to, the Property, including:

(a) in holding the mineral titles comprising the Property in good standing (including Maintenance Payments and Maintenance Filings), curing title defects and in acquiring and maintaining surface, water and other ancillary rights, all other

rentals, duties, assessments, payments, fees and other governmental charges applicable to, or imposed on, the Property, or in connection with holding the mineral titles comprising the Property, and all duties and Taxes levied against or in respect of the Property, and for activities on the Property;

- (b) in preparing for and in the application for and acquisition of environmental and other permits necessary or desirable to commence and complete exploration and development activities;
- in connection with any applications and necessary studies for the obtaining of permits, licences, and other regulatory approvals, including the preparation for and attendance at hearings and other meetings relating to the Property;
- (d) in doing geophysical, geochemical and geological surveys, drilling, assaying and metallurgical testing, including costs of assays, metallurgical testing and other tests and analyses (including downhole photography) to determine the quantity and quality of Minerals, water and other materials or substances;
- (e) in the preparation of Work Programs and Budgets and the presentation and reporting of data and other results obtained from those Work Programs, including any program for the preparation of reports or studies, including any preliminary economic assessment, scoping study, technical report, pre-feasibility study, feasibility study or other evaluation of the Property;
- (f) in searching for, digging, trenching, sampling, assaying, testing, working, developing, mining or extracting Minerals;
- (g) in conducting the drilling of holes by any method;
- (h) in acquiring, erecting and installing a mining plant, milling and metallurgical plant, ancillary facilities, buildings (including accommodations for workers, if necessary), machinery, tools, appliances or equipment and constructing access roads, railroads and other transportation facilities and, if necessary, water pipelines for use in relation to the Property;
- (i) in transporting Minerals, personnel, supplies, mining or milling plant, buildings, machinery, tools, appliances or equipment in, to or from the Property;
- (j) for environmental remediation and rehabilitation;
- (k) in acquiring or obtaining the use of facilities, equipment or machinery, and for all parts, supplies and consumables;
- for salaries, wages and/or other expenses for Persons assigned to conduct Operations (including reasonable travel, food and lodging expenses and other reasonable needs of such Persons);

- (m) in paying assessments or contributions under worker's compensation, employment insurance, pension or other similar legislation or ordinances relating to such personnel;
- (n) payments to contractors or consultants for work done, services rendered or materials supplied;
- (o) the costs of insurance premiums and performance bonds or other security;
- (p) in obtaining independent legal services directly relating to Operations; and
- (q) all Acquisition Costs incurred upon the acquisition of Additional Mineral Rights and Additional Other Rights pursuant to Article 6,

and, for clarity, a Party may, subject to the terms of this Agreement, either directly or through one of its Affiliates, provide the foregoing goods or services provided that the costs of such goods or services shall be charged at rates no higher than those which would be used by a non-related party in a transaction at arm's length for equivalent goods or services.

"First Share Issuance" has the meaning set out in Section 3.1(b).

"Governmental Authorities" means any foreign, domestic, national, federal, provincial, territorial, state, regional, municipal or local government or authority, quasi government authority, fiscal or judicial body, government or self regulatory organization, commission, board, tribunal, organization, or any regulatory, administrative or other agency, or any political or other subdivision, department, or branch of any of the foregoing.

"**Interest**" means, in the case of the Vendor, a 100% registered and beneficial interest in and to the Property and the rights of the Vendor pursuant to this Agreement, and in the case of the Purchaser, the Option, and the rights of the Purchaser pursuant to this Agreement.

"**IPO**" means an initial public offering or such other transaction, whether by means of take-over, merger, amalgamation, arrangement, reorganization, exchange of assets or a similar transaction or transactions, whereby, upon completion thereof, the Common Shares are listed on an Exchange.

"**IPO Price**" means the issue price (or deemed issued price) of Common Shares in connection with the IPO, subject to any applicable rules of the Exchange.

"List" has the meaning set out in Section 11.2(e).

"Maintenance Filings" has the meaning set out in Section 5.6.

"Maintenance Payments" has the meaning set out in Section 5.6.

"**Market Price**" means the volume weighted average trading price of the Common Shares on the Exchange for the five days on which such Common Shares were traded immediately preceding the date in respect of which the Market Price is to be determined, subject to any applicable rules of the Exchange.

"**Minerals**" means all marketable naturally occurring metallic minerals or mineral bearing material in whatever form or state, including, without limitation, any precious metal, any base metal or diamonds mined, extracted, removed, produced or otherwise recovered from the Property (but, for greater certainty, not including any rock, sand, gravel or aggregate), whether in the form of ore, doré, concentrates, refined metals or any other beneficiated or derivative products thereof and including any such metallic minerals or mineral bearing materials or products derived from any processing or reprocessing of any tailings or other waste products originally derived from the Property.

"**Miscellaneous Interests**" means the interests of the Vendor in all property, assets and rights (other than the Property) ancillary to the Property to which the Vendor is entitled including, but not limited to, the interests of the Vendor in:

- (a) rights under all Contracts and documents relating to the Property and the Operations conducted thereunder (but for greater certainty does not require the Purchaser to assume any liability in respect of such Contracts and documents);
- (b) all subsisting rights to enter upon, use and occupy the surface of any lands forming part of the Property or of any lands to be traversed in order to gain access to any of the lands forming part of the Property;
- (c) any right, license, concession or permit in relation to the use or diversion of water;
- (d) all permits, licenses and authorizations relating to the Property;
- (e) all books, records, data (including technical data) and other information relating to the Property, including accounting records, plans, drawings, specifications and other Operations records;
- (f) all pre-paid expenses and deposits relating to the Property, including all pre-paid taxes, rentals, licence fees and water rates, as well as pre-paid purchases of gas, oil and hydro;
- (g) all right, title, benefit and interest of the Vendor in and to all of the patents, trademarks, copyrights, designs, inventions, licences, sub-licences, processes, technology and other industrial property and intellectual property of or used in connection with the Property; and
- (h) any Additional Other Rights acquired pursuant to this Agreement.

"Notice" has the meaning set out in Section 11.9.

"**Operations**" includes every kind of work done on or in respect of the Property or the products therefrom and, without limiting the generality of the foregoing, includes the work of assessment, geophysical, geochemical and geological surveys, studies and mapping, investigating, drilling,

designing, examining, equipping, improving, surveying, shaft sinking, raising, cross-cutting and drifting, searching for, digging, trucking, sampling, working and procuring Minerals, surveying and bringing any mining claims to lease or patent, and doing all other work usually considered to be prospecting, exploration, development and/or mining work.

"Option" has the meaning set out in Section 3.1.

"Option Conditions" has the meaning set out in Section 3.1.

"Option Expenditures" has the meaning set out in Section 3.1(c).

"Option Period" means the period of time from the Effective Date to the date upon which the Purchaser exercises the Option or the Option terminates, all pursuant to the terms hereof.

"Parties" and "Party" have the meanings set out in the recitals to this Agreement.

"Permitted Encumbrances" means:

- rights of way for, or reservations or rights of others relating to, sewers, water lines, gas lines, pipelines, electric lines, telegraph and telephone lines and other similar products or services;
- (b) any reservations or exceptions to or vested in any Governmental Authority by the terms of any lease, licence, grant or permit forming part of the Property as of the date of this Agreement;
- (c) any encumbrance right or royalty vested in favour of any Governmental Authority arising under Applicable Laws or under the terms of any contract, mineral concession or license as of the date hereof;
- (d) security given to a public utility or any Governmental Authority when required in the ordinary course of business in connection with the Operations; and
- (e) any other rights or Encumbrances consented to in writing by the Purchaser (in its sole and absolute discretion) or granted by the Purchaser.

"**Person**" means any individual, sole proprietorship, partnership, unincorporated association, unincorporated syndicate, unincorporated organization, trust, company, corporation or other body corporate, union, Governmental Authority and a natural person in his capacity as trustee, executor, administrator, or other legal representative.

"**Property**" means the mineral titles depicted in Schedule A and any Additional Mineral Rights acquired pursuant to this Agreement.

"Purchaser" has the meaning set out in the recitals to this Agreement.

"Responding Party" has the meaning set out in Section 11.2(c).

"Response Period" has the meaning set out in Section 11.1(d).

"**Royalty**" means a two percent (2%) net smelter returns royalty having the terms set forth in Schedule "B", which shall be granted to Vendor, upon exercise of the Option, all in accordance with, and subject to the reduction contained in Section 4.2(b).

"Royalty Reduction Payment" has the meaning set out in Section 4.2(b).

"Rules" has the meaning set out in Section 11.2(e).

"Securities Act" means the Securities Act (British Columbia) and the rules, regulations, forms and published instruments, policies, bulletins and notices made thereunder, as now in effect and as they may be promulgated or amended from time to time.

"Share Issuances" has the meaning set out in Section 3.1(b).

"**Substance**" means any contaminant, pollutant or hazardous substance that is likely to cause harm or degradation to the environment or risk to human health or safety, including any pollutant, contaminant, waste, hazardous waste, toxic substance or dangerous good which is defined or identified in any Environmental Law.

"**Successors**" means successors and includes any successor continuing by reason of amalgamation or other reorganization and any Person to which assets are transferred by reason of a liquidation, dissolution or winding-up.

"Tax" means all federal, state, provincial, territorial, regional, county, municipal, local or foreign taxes, duties, imposts, levies, assessments, tariffs and other charges imposed, assessed or collected by a Governmental Authority, including:

- (a) any gross income, net income, gross receipts, business, royalty, capital, capital gains, goods and services, value added, severance, stamp, franchise, occupation, premium, capital stock, sales and use, real property, land transfer, personal property, ad valorem, transfer, licence, profits, windfall profits, environmental, payroll, employment, employer health, pension plan, antidumping, countervail, customs or excise tax;
- (b) all withholdings on amounts paid to or by the relevant Person;
- (c) all employment insurance premiums, government pension plan contributions or premiums;
- (d) any fine, penalty, interest, or addition to tax;
- (e) any tax imposed, assessed, or collected or payable pursuant to any tax-sharing agreement or any other contract relating to the sharing or payment of any such tax, levy, assessment, tariff, duty, deficiency, or fee; and
- (f) any liability for any of the foregoing as a transferee, successor, guarantor, or by contract or by operation of Applicable Law.

"Tax Act" means the Income Tax Act (Canada) as amended from time to time.

"**Third Party**" means a Person or other form of enterprise that is not a Party or an Affiliate of a Party.

"Transfer" means to sell, transfer, grant, assign, donate, create an Encumbrance, grant a right to purchase or in any other manner convey, transfer, alienate or dispose of, or commit to do any of the foregoing.

1.2 <u>Schedules</u>

The following Schedules to this Agreement, as listed below, are an integral part of this Agreement:

ScheduleDescriptionSchedule ADescription of the PropertySchedule BRoyalty Agreement

ARTICLE 2 REPRESENTATIONS AND WARRANTIES

2.1 <u>Vendor Representations and Warranties</u>

The Vendor represents and warrants to the Purchaser as follows:

- (a) the Vendor is the registered and beneficial owner of a 100% undivided interest in and to the Property, free and clear of all Encumbrances, Claims and defects in title, other than the Permitted Encumbrances;
- (b) the mineral titles comprising the Property have been duly and validly issued and acquired pursuant to all Applicable Laws in Ontario, are currently in good standing up to and including the expiry dates set forth in Schedule A hereto and are in full compliance with Applicable Laws and the depiction and description of the Property set forth herein (including in Schedule A) is true and correct;
- (c) all Operations conducted on the Property to the date of this Agreement by or on behalf of the Vendor, and to the knowledge of the Vendor, by or on behalf of any previous owner of the Property or any Person who had an option or interest in respect of the Property, have been conducted in all material respects in accordance with all Applicable Laws and conditions on and relating to the Property are in compliance with Applicable Laws, and no condition exists or event has occurred which, with or without notice or the passage of time or both, would constitute a violation of or give rise to material liability under any Applicable Law;
- (d) there has been no spill, discharge, leak, emission, ejection, escape, dumping, or any release or threatened release of any kind, of any Substance from, on, in or under the Property or into the environment, nor has any Substance or waste been treated on or disposed of, or is located or stored on the Property, as a result of the activities by or on behalf of the Vendor, or to the knowledge of the

Vendor, by or on behalf of any previous owner of the Property or any Person who had an option or interest in respect of the Property, except if expressly permitted by, and in compliance with, Applicable Law;

- (e) the Vendor has all required approvals and authorizations to grant the Option to the Purchaser, and to transfer a 100% interest in the Property and Miscellaneous Interests to the Purchaser in accordance with the terms hereof;
- (f) except for the Permitted Encumbrances and the rights of the Purchaser under this Agreement, the Vendor has not done any act or suffered or permitted any action to be done whereby any Person may acquire any interest in or to (or otherwise deal with) the Property or Miscellaneous Interests or in the Minerals or other products to be produced or removed from the Property, including, without limitation, any royalty or payment in the nature of rent;
- (g) no Person has any right under proprietary, possessory, preferential, pre-emptive or first purchase rights or otherwise to acquire any interest in or to (or otherwise deal with) the Property or Miscellaneous Interests or Minerals or other products to be produced or removed from the Property, or that might be triggered by virtue of this Agreement or the transactions contemplated hereby;
- (h) there is no actual, threatened or, to the knowledge of the Vendor, contemplated Claim or challenge relating to the Property, nor to the knowledge of the Vendor, is there any basis therefor, and there is not presently outstanding against the Vendor, nor to the knowledge of the Vendor, against any previous owner of the Property or any Person who had an option or interest in respect of the Property, any judgment, decree, injunction, rule or order of any court, Governmental Authority or arbitrator which would have an effect upon the Property;
- subject to any limitations imposed by Applicable Law, the Vendor has access to the Property to enable it to explore for Minerals and develop a mining project thereon, and the terms of the mineral titles comprising the Property and Applicable Law permit full, legal, binding and valid consent to access the surface area covered by the Property and to carry out all actions contemplated under the scope of this Agreement (including, but not limited to, conducting Operations on the Property);
- (j) the Property does not lie within any legacy claim, crown grant, privately held mineral rights, protected area, rescued area, reserve, reservation, reserved area, environmental or historic protected area as designated by any Governmental Authority having jurisdiction, that would materially and adversely impair the exploration for Minerals or other Operations on the Property, and the Property does not lie within any other lands in which mineral rights cannot be acquired;
- (k) there are no Claims, asserted, existing or pending made by or on behalf of any Aboriginal Peoples relevant to the Property or any lands included in the area of the Property or the Vendor (other than the general claim made by the Aboriginal Peoples concerning their traditional land) nor to the knowledge of the Vendor, is there any basis therefor;

- (I) neither the Vendor, nor to the knowledge of the Vendor, any previous owner of the Property or any Person who had an option or interest in respect of the Property, has notice, or knowledge of, any proposal to terminate or vary the terms of, or rights attaching to, the mineral titles comprising the Property from any Governmental Authority, of any challenge to the Vendor's right, title or interest in the Property or of any actual or alleged breach of any Applicable Laws, and there are no orders, directions or actions relating to environmental matters requiring any work, repairs, construction or capital expenditures with respect to the Property or the conduct of the business, including any Operations, related to the Property;
- (m) all work or expenditure obligations applicable to the Property, all statements and reports of the work or expenditures all rentals, duties, taxes, assessments, payments, fees and other governmental charges applicable to, or imposed on, the mineral titles comprising the Property and any other requirements to be paid, satisfied or filed to keep the mineral titles comprising the Property in good standing have been paid, satisfied or filed and have satisfied the applicable Governmental Authority;
- (n) the Vendor is not a party to or bound by any guarantee, indemnification, surety or similar obligation pertaining to the Property and, except for this Agreement, no Contracts have been entered between or on behalf of the Vendor and/or any other Person with respect to (or that may affect) the Property;
- (o) the Vendor is a resident of Canada for the purposes of the *Tax Act*; and
- (p) the Vendor has made available to the Purchaser all material maps, assays, surveys, drill logs, samples, metallurgical, geological, geophysical, geochemical and engineering data and other Operations records within its control (or the control of an Affiliate or any Person that conducted Operations on behalf of the Vendor) in respect of the Property and all information supplied to the Purchaser or its advisors or its personnel in the course of the due diligence review in respect of the transactions contemplated by this Agreement, is to the knowledge of the Vendor, accurate and correct in all material aspects and does not contain any untrue statement of material fact or omit to state any material fact necessary in order to make the statements contained therein not misleading.

2.2 <u>Mutual Representations and Warranties</u>

The Purchaser represents and warrants to the Vendor and Vendor represents and warrants to the Purchaser, as follows:

(a) in the case of the Purchaser, it is duly organized and validly existing under the laws of the place of its establishment or incorporation and is in good standing and existing under the laws of the place of its establishment or incorporation, and in the case of the Vendor, he is the full age of majority and is legally competent to execute this Agreement and take all action pursuant hereto;

- (b) this Agreement constitutes a legal, valid and binding obligation of the Party, enforceable against it in accordance with its terms by appropriate legal remedy, subject, however, to limitations with respect to enforcement imposed by Applicable Law in connection with bankruptcy or similar proceedings and to the extent that equitable remedies such as specific performance and injunction are in the discretion of the court from which they are sought;
- (c) the Party is not insolvent under the Applicable Laws of its jurisdiction and is able to pay its debts as they fall due;
- (d) neither the execution and delivery of this Agreement, nor the performance of its obligations hereunder or thereunder, nor the consummation of the transactions hereby contemplated conflict with any of, or require the consent or waiver of rights of any Person, nor do or shall do any of the foregoing:
 - (i) violate any provision of or require any consent, authorization or approval under any Applicable Law;
 - (ii) conflict with, result in a breach of, constitute a default under (whether with notice or the lapse of time or both), accelerate or permit the acceleration of the performance required by, or require any consent, authorization or approval which has not been obtained under, any contract to which it the Party is a party or by which the Party is bound or to which any of the Party's property is subject; or
 - (iii) result in the creation of any Encumbrance on the Property, except in accordance with the terms of this Agreement;
- (e) the Party has all requisite power and authority required to enter into this Agreement and each other document or instrument delivered in connection herewith and has all requisite power and authority to perform fully each and every one of its obligations hereunder; and
- (f) the Party has not incurred any liability, contingent or otherwise, for brokers' or finders' fees in respect of the transactions contemplated herein.

2.3 <u>Survival and Investigation</u>

- (a) The representations and warranties contained in this Agreement are conditions on which the Parties have relied in entering into this Agreement, and enforcement of such representations and warranties shall survive the execution hereof for a period of two years following the date of termination of this Agreement pursuant to Section 8.1, to the full extent necessary for the protection of the Party in whose favour they run.
- (b) Each Party shall indemnify and save the other harmless from all loss, damage, costs, actions and suits arising out of or in connection with any breach of any representation, warranty, covenant, agreement or condition made or to be fulfilled by it hereunder. In addition, the Vendor assumes, and releases the

Purchaser from, all existing obligations and liabilities as of the Effective Date, howsoever arising relating to the Property, and indemnifies and saves harmless the Purchaser from and against all Claims relating thereto, made against the Purchaser, any Affiliates thereof and any of their respective officers, directors, employees and representatives with respect to any existing obligations and liabilities as of the Effective Date, howsoever arising, relating to the Property.

- (c) A Party may waive any of such representations, warranties, covenants, agreements or conditions in whole or in part at any time without prejudice to its right in respect of any other breach of the same or any other representation, warranty, covenant, agreement or condition.
- (d) The right of any of the Parties to enforce any breach (and any remedy as a result of such enforcement) of any of the representations and warranties set out in Article 2 shall not be affected by any investigation conducted, or any knowledge acquired, at any time by the Party enforcing such breach, whether before or after the execution and delivery of this Agreement with respect to the accuracy or inaccuracy of, or compliance with, such representation or warranty.
- (e) In no event shall any Party be liable for any consequential damages or any damages based on loss of revenue or loss of profit in respect of any breach of any representation, warranty, covenant, or agreement set out in this Agreement.

ARTICLE 3 OPTION

3.1 Grant of Option

The Vendor hereby grants to the Purchaser the sole and exclusive right and option (the "**Option**") exercisable in the manner described herein, to acquire a 100% legal and beneficial interest in and to the Property free and clear of all Encumbrances and Claims, other than the Permitted Encumbrances and the Royalty, which interest shall be deemed to vest and be fully exercised on the date (the "**Acquisition Date**") upon which each of following conditions has been satisfied:

(a) <u>Cash Payments:</u> the Purchaser making cash payments to the Vendor as follows:

DATE	AMOUNT
Within 7 business days of execution of the Option Agreement (the " Effective Date ")	\$40,000
On or before the first anniversary of the Effective Date	\$40,000
On or before the second anniversary of the Effective Date	\$50,000
TOTAL	\$130,000

(the "Cash Payments")

and

(b) <u>Share Issuances</u>: the Purchaser (or resulting issuer, as applicable) issuing Common Shares to the Vendor as follows:

DATE	AMOUNT
As soon as practicable following completion of an IPO, and in any event within 10 Business Days of the IPO (subject to the below)	Such number of Common Shares as is equal to:
Notwithstanding the foregoing, if an IPO is not completed within 180 days of the Effective Date, at the Purchaser's sole option, it may elect to satisfy the First Share Issuance by making a cash payment to the Vendor in the amount of \$40,000 by no later than the date that is 190 days following the Effective Date	\$40,000 / the IPO Price (the " First Share Issuance ")
On or before the first anniversary of the Effective Date	Such number of Common Shares as is equal to:
	\$40,000 / Market Price
On or before the second anniversary of the Effective Date	Such number of Common Shares as is equal to:
	\$50,000 / Market Price

(the "Share Issuances")

and

(c) <u>Expenditures</u>: the Purchaser funding or incurring Expenditures as follows:

DATE	AMOUNT
On or before the first anniversary of the Effective Date	\$100,000
On or before the second anniversary of the	\$250,000

Effective Date	
On or before the third anniversary of the Effective Date	\$500,000
TOTAL	\$850,000

(the "**Option Expenditures**" and together with the Cash Payments and the Share Issuances, collectively, the "**Option Conditions**").

On the Acquisition Date, the Vendor shall take all actions and do all things necessary or desirable to effect a transfer of 100% of the legal and beneficial right, title and interest in and to the Property to the Purchaser in accordance with Section 4.1.

3.2 <u>No Obligation</u>

The Purchaser has the right, but not the obligation, to make the Cash Payments, complete the Share Issuances and fund or incur the Option Expenditures and do all other things necessary in order to exercise the Option pursuant to this Article 3. The Purchaser shall have the right at any time to elect not to continue to maintain the Option and to terminate this Agreement.

3.3 Cash in Lieu and Excess Expenditures

- (a) Subject to Section 3.3(b), the Purchaser may, in its sole discretion, make up any shortfall in the Option Expenditures required to be funded by the Purchaser by the end of the periods set out in Section 3.1(c) by making a cash payment to the Vendor before the end of the applicable period set out in Section 3.1(c), and the payment of such shortfall shall be deemed to have satisfied such requirement for the completion of the Option Expenditures during such period. The amount of any such shortfall so paid to the Purchaser shall be forwarded to and held by the Operator and be used for future Option Expenditures, and may not be distributed to any of the Parties.
- (b) If the amount of Expenditures actually funded by the Purchaser in a given period set out in Section 3.1(c) is below the minimum required Option Expenditures to be funded in such period because (i) the work performed by the Operator pursuant to an approved Work Program and Budget cost less than the Budget, (ii) the Operator did not, or was not able to (for any reason, including, without limitation, as a result of default or non-performance by the Operator of its obligations under this Agreement) complete the required Expenditures set out in an approved Work Program and Budget within such period, (iii) the Operator did not submit to the Purchaser cash calls equal to the required Expenditures set out in an approved Work Program and Budget within such period; and/or (iv) the Operator did not submit to the Purchaser (for the Purchaser's approval) a Work Program and Budget that set out the Option Expenditures required for such period, then:

- (A) the minimum required Option Expenditures to be funded in such period shall be deemed to be reduced to the amount actually funded by the Purchaser during such period; and
- (B) the difference between the amount of Option Expenditures required to be funded in such period and the amount of Expenditures actually funded by the Purchaser during such period shall be added to the amount of Option Expenditures required to be funded by the Purchaser in the immediately succeeding 12 month period set out in Section 3.1(c).
- (c) Any excess Expenditures completed in advance of a period specified in Section 3.1(c) shall be carried over and shall qualify and be accounted for as Option Expenditures made by the Purchaser during a subsequent period specified in Section 3.1(c).

3.4 <u>Accelerated Expenditures</u>

Any First Option Conditions may be satisfied within a shorter time frame than set out in Section 3.1, at the sole and absolute discretion of the Purchaser.

3.5 <u>Share Issuance Matters</u>

- (a) The Share Issuances pursuant to Section 3.1(b) shall be subject to the Securities Act and the policies of the Exchange, including, without limitation, receipt of any required regulatory approvals in connection therewith.
- (b) The Vendor understands and acknowledges that the Common Shares issued to the Vendor pursuant to the Share Issuances may be subject to statutory restrictions on resale and trading, as well as escrow arrangements, pursuant to Applicable Law, including, without limitation, the Securities Act and the policies of the Exchange. The Vendor agrees that it will enter into any escrow arrangement as may be required by Applicable Law.

3.6 Notice of Option

The Purchaser shall have the right to register notice of this Agreement with the applicable Governmental Authority office for the sole purpose of giving notice of its Option rights hereunder. Such notice shall be removed by the Purchaser upon termination of this Agreement.

ARTICLE 4 VESTING OF INTEREST AND SUBSEQUENT MATTERS

4.1 <u>Vesting of 100% Interest</u>

On the Acquisition Date:

- (a) the Option is deemed fully exercised;
- (b) the Purchaser shall give Notice to Vendor of such fact; and

(c) the Vendor shall take all actions and do all things necessary or desirable to effect a transfer of 100% of the legal and beneficial right, title and interest in and to the Property and the Miscellaneous Interests to the Purchaser, such that the Purchaser thereafter holds a 100% legal and beneficial interest in and to the Property and the Miscellaneous Interests, free and clear of all Encumbrances and Claims, other than the Permitted Encumbrances and the Royalty.

4.2 <u>Royalty</u>

- (a) From and after transfer of 100% of legal and beneficial right, title and interest in and to the Property and the Miscellaneous Interests as set out in Section 4.1, the Vendor shall be entitled to be granted the Royalty upon the terms and conditions set out in Schedule "B".
- (b) The Purchaser shall at any time following the Acquisition Date be entitled, upon written notice to the Vendor, to purchase from the Vendor 1% of the Royalty in exchange for payment of \$500,000, thereby reducing the Royalty held by the Vendor to 1% (the "**Royalty Reduction Payment**").

ARTICLE 5 OPERATOR AND OPERATIONS

5.1 <u>Operator</u>

During the Option Period, subject to the provisions of this Article 5, the Vendor shall act as the operator of the Properties (the "**Operator**") and shall conduct the Operations, and the Purchaser shall be the "**Non-Operator**").

5.2 <u>General Duties of Operator</u>

In addition to the other obligations set forth herein, the Operator shall do the following:

- (a) perform, and cause its employees, consultants, sub-contractors and representatives to perform, the Operations in a prudent and workmanlike manner, with the degree of effort, skill and judgment that is in accordance with good Canadian exploration, construction, mining, processing and engineering practices generally and locally prevailing in the mining industry and in accordance with all Applicable Laws and all agreements, permits and licences relating to the Property and the Operator;
- (b) pay and discharge all wages and accounts for material and services and all other costs and expenses that may be incurred by the Operator in connection with its Operations on the Property, and to save the Purchaser harmless from and against all liens in respect of such Operations which may be filed against the Property, and in the event of any liens being so filed, to proceed forthwith to have the same removed, provided that the foregoing provision shall not prevent the Operator from properly contesting in good faith any claims for liens which the Operator considers unjustified;

- (c) indemnify and hold harmless the Purchaser, its directors, officers, employees, agents or representatives (the "Indemnitees") from and against all claims, losses, liabilities, demands, costs (including reasonable attorneys' fees and expenses incurred by the Purchaser), damages, actions, suits or other proceedings whatsoever arising out of or attributable to any fraud, negligence, wilful misconduct, or as a result of the breach of any Applicable Laws, committed by the Operator and/or its employees, consultants, sub-contractors and representatives;
- (d) apply for all necessary permits, licenses and approvals, comply with all Applicable Laws and promptly notify the Purchaser of any allegations of material violation thereof;
- (e) obtain and maintain (and cause its subcontractors to obtain and maintain), comprehensive general liability in such coverage amounts and on such terms as are consistent with industry standards and practices, and as approved by the Purchaser in its sole and absolute discretion;
- (f) maintain financial books and records showing all costs, Expenditures, receipts, and disbursements in connection with all Operations. These accounts shall include general ledgers and supporting and subsidiary journals, invoices, cheques, and other customary documentation;
- (g) deliver to the Purchaser, at the end of each quarter and of each calendar year (or in such other frequency as may be determined by the Purchaser, in consultation with the Purchaser), progress reports, along with an itemized report of the aggregate Expenditures spent by the Operator in carrying out the approved Work Program by the Operator during such period, certified to be correct by the Operator (each an "Expenditure Report"), in form satisfactory to the Purchaser. The progress reports shall indicate the status of any approved Work Program being conducted on the Property, and the applicable Budget, and disclose any significant technical data learned or obtained in connection with such work. Such reports shall be delivered within 30 days (or in such other timeframe as may be determined by the Purchaser, in consultation with the Operator) of the end of the quarter or calendar year to which such report relates;
- (h) perform its, and cause its employees, consultants, sub-contractors and representatives to perform their, activities under this Agreement, with good health and safety standards expected from recognized international operators experienced in providing the types of services contemplated in this Agreement in similar locations and in compliance with Applicable Laws regarding health and safety. The Operator shall be solely responsible for the health and safety relating to its activities, its employees, consultants, sub-contractors and representatives; and
- (i) provide the Purchaser and its representatives with access to the Property and to the books and records maintained by the Operator, during regular business hours, with or without advance notice.

5.3 <u>Removal of Operator</u>

The Purchaser shall have the option at any time, in its sole and absolute discretion, to: (a) provide written notice of termination to the Operator, designating a date of termination; and (b) appoint a successor Operator, which such successor Operator may be the Purchaser, an Affiliate of the Purchaser, or a Third Party. For greater certainty, the Purchaser's option to remove and replace the Operator is not contingent on the Operator being in default of any provisions of this Agreement.

5.4 <u>Work Programs and Budgets</u>

All Operations shall be conducted and all costs shall be incurred on the basis of an approved work program (a "**Work Program**"), and a budget ("**Budget**"), except in the case of emergency actions in accordance with Section 5.5. All Work Programs and Budgets shall be prepared by the Operator unless otherwise determined by the Purchaser, in its sole and absolute discretion. All proposed Work Programs and Budgets must be approved by the Purchaser, in its sole and absolute discretion, before implementation by the Operator, on at least an annual basis or more frequently as required, including any Work Programs and Budgets required to keep the Properties in good standing. Upon approval of a Work Program and Budget by the Purchaser, the Operator shall implement such Work Program and Budget within the time periods set out therein, and otherwise in compliance with its duties and obligations under this Agreement.

5.5 <u>Emergency or Unexpected Funding</u>

Notwithstanding any other provisions hereof, in case of emergency or to address unexpected events or to cover unexpected liabilities or expenses not covered in an approved Work Program and Budget which are necessary to protect against loss, injury or damage to Persons or property, or to protect the Operations and the Property or to comply with Applicable Laws, the Operator may take any reasonable action the Operator deems necessary and may incur such Expenditures as it deems necessary, notwithstanding that such Expenditures shall exceed allowable Expenditures under an approved Work Program and Budget. The Operator shall promptly notify the Purchaser of any such emergency or unexpected Expenditures that have been made or taken or that must be made or taken, and the funds necessary to pay for such emergency and unexpected Expenditures shall be added to the current Work Program and Budget and shall constitute Expenditures for the purposes of Section 3.1(c), subject however, to the indemnity requirements in Section 5.2(c), if applicable.

5.6 <u>Maintenance Payments</u>

The Operator covenants to, at all times, keep the Properties in good standing and to Property comply with Applicable Laws and the terms and conditions applicable to the Property, including by making, or causing to be made, as applicable, all filings, reports, applications (including renewal applications) (the "**Maintenance Filings**") within the time periods necessary, and making, or causing to be made, as applicable, the minimum required tax, Expenditures and other maintenance payments (the "**Maintenance Payments**") to keep the Property in good standing or as otherwise required by Applicable Laws or the terms and conditions applicable to the Property and must provide the Purchaser with written evidence of the Maintenance Filings and the Maintenance Payments, subsequent to filing or payment, respectively, thereof, and if the Operator should breach these covenants and any breach continues for a period of 14 days

following written notice by the Non-Operator to the Operator of such breach, then the Non-Operator shall be entitled (but under no circumstances obligated) to cure such default (a "**Maintenance Default**"). All costs incurred by the Non-Operator in curing such default shall constitute (and qualify as) Expenditures for the purposes of Section 3.1(c). In addition to the foregoing, upon request at any time by the Purchaser in its sole and absolute discretion, the Operator must provide the Purchaser with copies of (and the opportunity to review) any Maintenance Filings prior to being filed.

ARTICLE 6 AREA OF INTEREST

6.1 <u>Area of Interest</u>

For the purposes of this Agreement, the "**Area of Interest**" means the area within two kilometers of the external boundaries of the combined geographic area of the Property.

6.2 <u>Prohibition on Acquisitions</u>

During the Option Period, no Party shall directly or indirectly acquire any of the following rights in any property or mineral title or rights related thereto that is located wholly or partly within the Area of Interest, except in accordance with this Article 6:

- rights (including option rights) to or interests in any minerals, whether surface placer rights, subsurface rights, or of any other nature (the "Additional Mineral Rights"); or
- (b) rights to or interests in any real property, including, without limitation, any surface rights or any right, concession, authorization, license, permit, or interest in relation to the use or diversion of water that may affect the development of rights to minerals (the "Additional Other Rights").

For greater certainty, the Area of Interest and the provisions this Article 6 shall only apply during the Option Period.

6.3 <u>Notice and Election.</u>

If any Party acquires or proposes to acquire any Additional Mineral Rights or Additional Other Rights in the Area of Interest, it must give Notice to the other Party of such acquisition and the non-acquiring Party shall have the election whether to add such Additional Mineral Rights or Additional Other Rights to the Property or the Miscellaneous Rights, respectively (whether contained wholly or partly within the Area of Interest). The Notice provided by the acquiring Party must include a summary of the terms, if any, on which the Additional Mineral Rights or Additional Other Rights are being acquired (or have been acquired), a statement of all costs and expenses incurred by the acquiring Party in acquiring or staking or acquiring rights or permits to the Additional Mineral Rights or Additional Other Rights (the "**Acquisition Costs**"), and a copy of all title documents and, if applicable, contracts relating to the acquisition of the Additional Mineral Rights or Additional Other Rights. The non-acquiring Party must exercise its election within 45 days of receiving Notice of the acquisition. If the non-acquiring Party does not exercise its election to add such Additional Mineral Rights or Additional Other Rights to the Property or the Miscellaneous Rights, respectively, in accordance with this Section 6.3, the Additional Mineral Rights or Additional Other Rights shall not be added to the Property or the Miscellaneous Rights, respectively, and the acquiring Party shall be at liberty to develop or otherwise deal with such Additional Mineral Rights or Additional Other Rights for its own account, and, if such acquiring Party acquires such Additional Mineral Rights and Additional Other Rights for its own account, such Additional Mineral Rights or Additional Other Rights shall no longer be subject to this Article 6 and shall be excluded from the Area of Interest.

6.4 <u>Costs and Expenses</u>

- (a) If the acquiring Party of the Additional Mineral Rights and/or Additional Other Rights is the Vendor, and the Purchaser exercises its election to add the Additional Mineral Rights or Additional Other Rights to the Property, or the Miscellaneous Rights, respectively, the Purchaser shall reimburse the Vendor for all Acquisition Costs incurred by the Vendor, provided that the amount so reimbursed shall constitute Expenditures for the purposes of Section 3.1(c).
- (b) If the acquiring Party of the Additional Mineral Rights and/or Additional Other Rights is the Purchaser, and the Owners exercise their election to add the Additional Mineral Rights or Additional Other Rights to the Property, or the Miscellaneous Rights, respectively, the Acquisition Costs incurred by the Purchaser shall constitute Expenditures for the purposes of Section 3.1(c).

6.5 No Enlargement of Area of Interest

The Area of Interest shall not be enlarged by the addition of any Additional Mineral Rights or Additional Other Rights to the Property or the Miscellaneous Rights, respectively, pursuant to this Article 6.

ARTICLE 7 RESTRICTIONS ON TRANSFER

- 7.1 <u>Restriction on Transfer</u>
 - (a) During the Option Period, the Vendor may not directly or indirectly Transfer all or any part of its Interest without the prior written consent of the Purchaser, in its sole and absolute discretion.
 - (b) The Vendor acknowledges that damages alone would not be an adequate remedy in the event that it Transfers (or permits the Transfer of) its Interest in contravention of this Section 7.1. Accordingly, without prejudice to any other rights and remedies that the Purchaser may have, the Purchaser shall be entitled to the granting of equitable relief (including without limitation injunctive relief and specific performance, without the requirement of posting a bond or other security) in connection with or arising out of any actual or threatened Transfer of the Interest in contravention of this Section 7.1.

ARTICLE 8 TERMINATION AND SURVIVAL

8.1 <u>Termination</u>

- (a) This Agreement shall be terminated upon the occurrence of any of the events set forth below:
 - (i) if, at any time, the Purchaser provides Notice to the Vendor of its decision not to advance with the Property;
 - (ii) upon the Purchaser exercising the Option in accordance with the terms hereof; or
 - (iii) subject to Sections 3.3, 3.4 and 9.1, in the event that the Purchaser is in default of its obligations to make any of the cash payments or share issuances or fund or incur any of the Expenditures within the applicable time periods therefor prescribed by Article 3 and provided that Notice of any such default has been provided in writing by the Vendor to the Purchaser, upon the Purchaser's failure to cure such default within 30 days of its receipt of such Notice.
- (b) In the event of termination pursuant to Sections 8.1(a)(i) or 8.1(a)(iii) above:
 - (i) the Purchaser shall acquire no interest in the Property;
 - (ii) the Operator shall deliver to the Purchaser, copies of all records, information and data in respect of the Property that existed between the Effective Date and the date of termination of this Agreement, and the Purchaser may maintain such copies use the information contained therein, subject to the terms of Article 10.
 - (iii) the Operator shall return to the Purchaser all monies contributed by the Purchaser in respect of a Work Program or Budget that have not been spent as at the date immediately prior to the date of termination of this Agreement; and
 - (iv) without the need of any further confirmation or formality, the Purchaser shall be absolved of any requirement or obligation to fund or incur Expenditures or any other payments or amounts.

8.2 <u>Survival</u>

Section 2.3, Section 5.2(b), Section 5.2(c), Article 8, Article 10 and Article 11 and all limitations of liability and rights accrued prior to completion, termination, or expiration of this Agreement shall not merge on completion, termination, or expiration of this Agreement, but shall continue in full force and effect after any termination or expiration of this Agreement as shall any other provision of this Agreement which expressly or by implication from its nature is intended to survive the termination or expiration of this Agreement.

ARTICLE 9 EVENT OF FORCE MAJEURE

9.1 <u>Event of Force Majeure</u>

- (a) No right of a Party shall be affected, and no Party shall be liable under this Agreement or found in default, under this Agreement by the failure of such Party to meet any term or condition of this Agreement where such failure is caused by an Event of Force Majeure and, in such event, all times specified or provided for in this Agreement shall be extended by a period commensurate with the period during which the Event of Force Majeure causes such failure.
- (b) A Party affected by an Event of Force Majeure shall take all reasonable steps within its control to remedy the failure caused by such event, provided however, that nothing contained in this Section 9.1 shall require any Party to settle any labour or industrial dispute or to test the constitutionality of any Applicable Law enacted by any Legislature or Parliament of or within Canada.
- (c) Nothing in this 9.1 shall require a Party affected by an Event of Force Majeure to settle labour disputes or question the validity of any act of a Governmental Authority.
- (d) Any Party relying on the provisions of this Section 9.1 shall forthwith give Notice to the other Parties of the commencement of an Event of Force Majeure and of its end.

ARTICLE 10 CONFIDENTIALITY

10.1 <u>Confidentiality of Information</u>

All information provided to or received by the Parties hereunder shall be treated as Confidential (**"Confidential Information**"). During the Option Period, the Parties shall each solicit the consent of the other to the disclosure of Confidential Information in circumstances other than those set forth in Section 10.2 and such consent shall not be unreasonably withheld or delayed. For greater certainty, the Parties agree and acknowledge that once consent is granted by a Party hereunder with respect to the disclosure of any particular Confidential Information, or such Confidential Information is otherwise disclosed pursuant to any exemption set forth in Section 10.2 below, the provisions of this Section 10.1 requiring consent shall no longer be required with respect to any subsequent disclosure of the same Confidential Information which has previously been consented to and/or disclosed.

10.2 <u>Permitted Disclosure</u>

The consent required by Section 10.1 shall not be required with respect to any disclosure:

- (a) required to comply with any Applicable Laws, stock exchange rules or a regulatory authority having jurisdiction;
- (b) to a director, officer, employee or other representative of a Party;

- (c) to an Affiliate of a Party;
- (d) to a consultant, contractor or subcontractor of a Party that has a *bona fide* need to be informed;
- (e) to any Third Party to whom the disclosing Party may, if permitted by the terms of this Agreement, assign any of its rights under this Agreement; or
- (f) if the disclosing party is the Purchaser (i) to a bank, financial institution or investor from which the Purchaser is seeking equity or debt financing; (ii) in a prospectus, offering memorandum or other publicly filed document in connection with an IPO, and/or pursuant to which the Purchaser is seeking to obtain financing; or (iii) in continuous disclosure documents prepared pursuant to Applicable Laws.

10.3 Exception

The obligations with respect to Confidential Information and prohibitions against use under this Agreement shall not apply to information that the disclosing Party can show by reasonable documentary evidence or otherwise:

- (a) as of the Effective Date, was in the public domain;
- (b) prior to the time of receipt, was known to the disclosing Party;
- (c) was furnished to the disclosing Party by a Third Party who is not bound by a confidentiality agreement with, or by any obligation of confidence to, the other Parties;
- (d) was independently acquired or developed by the disclosing Party without use of, or reference to, the Confidential Information of the other Parties and without otherwise contravening the terms and provisions of this Agreement; or
- (e) after the Effective Date, was published or otherwise became part of the public domain (but only after, and only to the extent that, it is published or otherwise becomes part of the public domain through no breach by the disclosing Party of Article 10).

ARTICLE 11 GENERAL

11.1 <u>Rules of Interpretation</u>

In this Agreement and the Schedules to this Agreement:

- (a) time is of the essence in the performance of the Parties' respective obligations;
- (b) all references to money amounts are to Canadian currency;

- (c) the words "**herein**", "**hereof**" and "**hereunder**" and other words of similar import refer to this Agreement as a whole and not to any particular Article, Section or Schedule;
- (d) the singular of any term includes the plural and vice versa and the use of any term is equally applicable to any gender and where applicable to a body corporate;
- (e) the word "or" is not exclusive and the word "including" is not limiting (whether or not non-limiting language such as "without limitation" or "but not limited to" or other words of similar import are used with reference thereto);
- (f) where the phrase "**to the knowledge of**" or phrases of similar import are used in this Agreement, it shall be a requirement that the Person in respect of whom the phrase is used shall have made such due enquiries as a prudent business Person in comparable circumstances would make and that are reasonably necessary to enable such Person to make the statement or disclosure;
- (g) the descriptive headings of Articles and Sections are inserted solely for convenience of reference and are not intended as complete or accurate descriptions of content and shall not be used to interpret the provisions of this Agreement;
- (h) any reference to a corporate entity includes, and is also a reference to, any corporate entity that is a Successor to such entity;
- (i) the representations, warranties, covenants and agreements contained in this Agreement shall not merge and shall continue in full force and effect from and after the date hereof for the applicable period set out in this Agreement;
- (j) all Schedules attached to this Agreement form part of this Agreement; and
- (k) whenever any payment is to be made or any action under this Agreement is to be taken on a day other than a Business Day, such payment shall be made or action taken on the next Business Day following.

11.2 Arbitration

- (a) In the event of a dispute in relation to this Agreement, including, without limitation, the existence, validity, performance, breach or termination thereof, or any matter arising therefrom, the Parties will attempt to resolve amicably any such dispute by referral to successively higher level of the Parties' respective management (as applicable). If there is no resolution of the dispute by this means within thirty (30) days, then such dispute can be resolved by pursuant to this Section 11.2.
- (b) A Party to this Agreement may request that the dispute be resolved by binding arbitration, conducted in English through the British Columbia International

Commercial Arbitration Centre, in Vancouver, British Columbia. The seat of arbitration shall be Vancouver, British Columbia.

- (c) To request arbitration, the moving Party shall give written Notice to the other Party (a "**Responding Party**"), which Notice shall toll the running of any applicable limitations of actions under any Applicable Law or under this Agreement. Such Notice shall specify the nature of the allegation and issues in dispute, the amount or value involved (if applicable) and the remedy requested.
- (d) Within 20 days of the receipt of the Notice (the "Response Period"), the Responding Party shall answer the demand in writing, specifying the allegations and issues that are disputed, and advising of any cross-claim or counterclaim that they intend to pursue in the arbitration. Any Party failing to respond in accordance with this Section shall not be entitled to further notice of, or participation in, the arbitral proceeding.
- (e) The arbitration shall be conducted a single arbitrator. The procedure for selecting the single arbitrator shall be as follows: each Party shall select an individual designee (the "Designee"), which such individual must be indicated as a panelist, arbitrator or mediator on the list of same maintained by the British Columbia International Commercial Arbitration Centre (the "List"). The two Designees so appointed shall, within 15 days of the expiry of the Response Period, mutually agree upon the arbitrator, who shall be a fellow panelist, arbitrator or mediator that is indicated as such on the List. If a Party fails to appoint a Designee within the Response Period, then the Party that did appoint a Designee within such period shall be entitled to select the arbitrator, which such arbitrator or mediator on the List. The arbitration shall be conducted in accordance with the British Columbia International Commercial Arbitration Centre's rules (the "Rules").
- The arbitrator shall fix a time and place in Vancouver, British Columbia (f) reasonably convenient for the Parties, after giving the Parties not less than seven (7) Business Days' notice, for the purpose of hearing the evidence and representations of the Parties and the arbitrator shall preside over the arbitration and determine all questions of procedure not provided for under the Rules or this Section. After hearing any evidence and representations that the Parties may submit, the arbitrator shall make a decision and reduce the same to writing and deliver one copy thereof to the Claimant and the Respondent. The arbitrator shall endeavor to make a decision within forty-five (45) days after its appointment, subject to any reasonable delay due to unavoidable circumstances. Any decision by the arbitrator shall follow and apply the laws applicable to this Agreement pursuant to Section 11.7. The expense of the arbitration, including travel costs, expert witness and legal fees and costs shall be paid as determined in the discretion of the arbitrator, having due regard for the outcome of the arbitration and the relationship of the result to the positions taken by the Parties. In the absence of fraud or manifest error, the decision of the arbitrator shall be final and binding upon each of the Parties and the Parties expressly exclude any and all

rights to appeal, set aside or challenge any award by the arbitrator insofar as such exclusion can be validly made.

- (g) Judgment upon the award may be entered by any court having jurisdiction thereof or having jurisdiction over the relevant Party or its assets. Except where matters are expressed herein to be subject to arbitration, the provincial or federal courts sitting in British Columbia, Canada shall have exclusive jurisdiction to hear and determine all matters relating to this Agreement, including enforcement of the obligation to arbitrate.
- (h) Nothing in this Agreement shall prevent any Party from applying to the provincial or federal courts in British Columbia, Canada for interlocutory, injunctive, provisional, or interim measures, including but not limited to any claim for preliminary injunctive relief.
- (i) A dispute of the Parties shall not constitute an Event of Force Majeure.
- (j) All papers, Notice or process pertaining to an arbitration hereunder may be served on a Party as provided in Section 11.9.
- (k) The Parties agree to treat as Confidential Information, in accordance with the provisions of Article 10, the following: the existence of the arbitral proceedings; written Notices, pleadings and correspondence in relation to the arbitration; reports, summaries, witness statements and other documents prepared in respect of the arbitration; documents exchanged for purposes of the arbitration; the contents of any award or ruling made in respect of the arbitration. Notwithstanding the foregoing part of this Section 11.2(j), a Party may disclose such Confidential Information in judicial proceedings to enforce, nullify, modify or correct an award or ruling and as permitted under Article 10, or as required by Applicable Law or as compelled by a lawful authority.

11.3 Entire Agreement

This Agreement, including the Schedules to this Agreement, together with the agreements and other documents to be delivered pursuant to this Agreement, constitute the entire agreement between the Parties pertaining to the subject matter hereof and supersede all prior agreements, understandings, negotiations and discussions, whether oral or written, of the Parties (including, without limitation, the Letter of Intent between the Parties dated July 24, 2020) and there are no warranties, representations or other agreements between the Parties in connection with the subject matter hereof except as specifically set forth in this Agreement and in any agreement or document delivered pursuant to this Agreement. No supplement, modification or waiver or termination of this Agreement shall be binding unless executed in writing by the Party to be bound thereby.

11.4 Parties' Rights to Conduct Other Business

Each Party shall devote such time as may be required to fulfill any obligation assumed by it hereunder but, except as otherwise provided in this Agreement:

- (a) outside of the Property and the Area of Interest, the Parties and their respective Affiliates shall be free to engage in any business or other activity, whether or not competitive with the activities of the other Parties, and whether or not such business activity or acquisition is a result of reviewing the information obtained from the Property, and in particular, this Agreement may not be construed to prevent a Party from acquiring any mineral rights or interests therein, real property rights, water rights, or other associated rights outside of the boundaries of the Property;
- (b) no Party shall be under any fiduciary or other obligation to any other Party which shall prevent or impede such Party from participating in, or enjoying the benefits of, competing endeavours of a nature similar to the business or activity undertaken by the Parties hereunder; and
- (c) the legal doctrines of "corporate opportunity" or "business opportunity" sometimes applied to Persons occupying a relationship similar to that of the Parties shall not apply with respect to participation by any Party in any business activity or endeavour outside the Property, and, without implied limitation, a Party shall not be accountable to the others for participation in any such business activity or endeavour outside the Property which is in direct competition with the business or activity undertaken by the Parties under this Agreement.

11.5 No Reliance or Inducement

Each Party represents and warrants and agrees that when entering into this Agreement it relied exclusively on the following matters independently of any statements, inducements or representations made by or on behalf of any other Party (including without limitation by any agents acting on behalf of a Party):

- (a) its own inspections, investigations, skill and judgement;
- (b) the terms expressly contained in this Agreement; and
- (c) opinions and advice obtained independently of any other Party.

11.6 <u>Tax Included and Refundable Tax Credits</u>

- (a) All amounts and Expenditures to be funded, paid or incurred by the Purchaser as described in this Agreement are inclusive of any amounts on account of Tax incurred by any Party in funding an approved Work Program (i.e. no tax gross up shall be made). Any such Tax recovered or refunded by either of the Parties during the term of this Agreement shall be used to fund Expenditures; however such recovered or refunded Tax shall not be credited towards the Purchaser's Option Expenditure obligations. If any Tax is recovered by the Vendor following termination of this Agreement, such Tax shall be reimbursed to the Purchaser.
- (b) Any Canadian exploration expenses that may be accumulated pursuant to the Tax Act that are derived from eligible Expenditures under the Tax Act, shall be for the account of the Purchaser and for its sole benefit.

11.7 <u>Applicable Law</u>

- (a) This Agreement shall be governed by and interpreted in accordance with the laws in force in the Province of British Columbia and the federal laws of Canada applicable therein, without regard to any conflict of laws or choice of laws principle that would permit or require the application of the laws of any other jurisdiction.
- (b) Each of the Parties hereby irrevocably attorns and submits to the arbitral jurisdiction set forth in Section 11.7(a) and, with respect to any matters not determined by arbitration, to the exclusive jurisdiction of the courts of British Columbia, Canada respecting all matters relating to this Agreement and the rights and obligations of the Parties hereunder.

11.8 Expenses

Except as otherwise provided, all costs incurred in connection with this Agreement and the transactions contemplated hereby shall be paid by the Party incurring them.

11.9 <u>Notices</u>

Any notice or writing required or permitted to be given under this Agreement or any communication otherwise made in respect of this Agreement (referred to in this Section as a "**Notice**") shall be sufficiently given if in writing and: (a) delivered personally, either to the individual designated below for such Party, or to an individual having apparent authority to accept deliveries on behalf of such individual at the address set out below for such Party; (b) by registered mail to the address set out below for such Party; or (c) transmitted by email, at or to the applicable email addresses set out below for such Party:

(i) In the case of Notice to the Vendor, at:

with a copy which shall not constitute notice to:

(ii) In the case of Notice to the Purchaser, at:

Notice provision redacted

with a copy which shall not constitute notice to:

Notice provision redacted

or at such other address as the Party to whom such Notice is to be given shall have last notified the Party giving the same, in the manner provided in this Section. Any Notice delivered to the Party to whom it is addressed as provided in this Section shall be deemed to have been given and received on the day it is so delivered at such address, provided that if such day is not a Business Day then the Notice shall be deemed to have been given and received on the Business Day next following such day. Any Notice mailed to a Party to whom it is addressed as provided in this Section shall be deemed to have been given and received on the third Business Day following the date of mailing, except in the event of disruption of the postal services, in which event, Notice will be deemed to have been given and received will be deemed only when actually received. Any Notice transmitted by facsimile or email shall be deemed given and received on the first Business Day after its transmission.

11.10 Assignment and Successors

This Agreement is binding upon and shall enure to the benefit of the Parties and their respective Successors and permitted assignees. Notwithstanding any other provision hereof, the Parties agree and acknowledge that the Purchaser shall be entitled to assign its Interest hereunder to any Third Party provided that it provides Notice of such assignment to the Vendor and such transferee(s) agree in writing to be bound by the terms and conditions of this Agreement.

11.11 <u>Severability</u>

If any provision of this Agreement is void, illegal or unenforceable, it may be severed without affecting the enforceability of the other provisions in this Agreement.

11.12 Waiver and Amendment

A waiver of any right, power or remedy under this Agreement must be in writing signed by the Party granting it. A waiver is only effective in relation to the particular obligation or breach in respect of which it is given. It is not to be taken as an implied waiver of any other obligation or breach or as an implied waiver of that obligation or breach in relation to any other occasion. No modification, variation or amendment of this Agreement is of any force unless it is in writing and has been signed by each of the Parties.

11.13 Relationship of the Parties

Nothing in this Agreement or any other agreement relating to the development of the Property is to constitute or be deemed to constitute a partnership or joint venture for any purpose between the Parties or an agency of one Party to another Party, or in any way create a fiduciary duty of one Party to another Party.

11.14 Further Assurances

Subject to the terms and conditions of this Agreement, the Parties shall use all reasonable efforts to take, or cause to be taken, all actions, and to do, or cause to be done, all things necessary, proper or advisable under Applicable Laws to carry out all of their respective obligations under this Agreement and to consummate the transactions contemplated by this Agreement, and from time to time, without further consideration, each Party shall, at its own expense, execute and deliver such documents to any other Party as such Party may reasonably request in order to consummate the transactions contemplated by this Agreement. Each of the Parties agrees to take all such actions as are within its power to control, and to use reasonable commercial efforts to cause other actions to be taken which are not within its power to control, so as to ensure compliance with each of the conditions and covenants set forth in this Agreement which are for the benefit of the other Party.

[Remainder of page intentionally left blank.]

11.15 Execution in Counterparts and by Electronic Delivery

This Agreement may be executed in any number of counterparts and by the different Parties hereto on separate counterparts, each of which when so executed and delivered shall be an original, but all such counterparts together shall constitute one and the same instrument. Such counterparts may be delivered by regular post, courier or electronic mail.

IN WITNESS WHEREOF the Parties have hereunto duly executed this Option Agreement as of the date first written above.

Signature redacted

WITNESS

Signature redacted

ALEX PLESON

Signature redacted **1257590 B.C. Ltd.**

Per:

Name: Eli Dusenbury Title: Director, CEO and CFO

SCHEDULE A

DESCRIPTION OF THE PROPERTY

Township / Area	Tenure ID	Tenure Type	Registered and Beneficial Title Holder	Anniversary Date
BARBARA LAKE AREA	599765	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA,LAKE JEAN AREA	599764	Multi-cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599763	Single Cell Mining Claim	Alex Pleson	2022-07-19
LAKE JEAN AREA	599762	Single Cell Mining Claim	Alex Pleson	2022-07-19
LAKE JEAN AREA	599761	Single Cell Mining Claim	Alex Pleson	2022-07-19
LAKE JEAN AREA	599760	Single Cell Mining Claim	Alex Pleson	2022-07-19
LAKE JEAN AREA	599759	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599758	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA,LAKE JEAN AREA	599757	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA,LAKE JEAN AREA	599756	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA,LAKE JEAN AREA	599755	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599754	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599753	Single Cell Mining Claim	Alex Pleson	2022-07-19

Township / Area	Tenure ID	Tenure Type	Registered and Beneficial Title Holder	Anniversary Date
BARBARA LAKE AREA	599752	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA,LAKE JEAN AREA	599751	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599750	Single Cell Mining Claim	Alex Pleson	2022-07-19
LAKE JEAN AREA	599749	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599748	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599747	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA,LAKE JEAN AREA	599746	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599745	Multi-cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599744	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599743	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599742	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599741	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599740	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599739	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599738	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599737	Single Cell Mining Claim	Alex Pleson	2022-07-19

Township / Area	Tenure ID	Tenure Type	Registered and Beneficial Title Holder	Anniversary Date
BARBARA LAKE AREA	599736	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599735	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599734	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599733	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599732	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599731	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599730	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599729	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599728	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599727	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599726	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599725	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599724	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599723	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599722	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599721	Single Cell Mining Claim	Alex Pleson	2022-07-19

Township / Area	Tenure ID	Tenure Type	Registered and Beneficial Title Holder	Anniversary Date
BARBARA LAKE AREA	599720	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599719	Multi-cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	596221	Single Cell Mining Claim	Alex Pleson	2022-06-17
BARBARA LAKE AREA	596220	Single Cell Mining Claim	Alex Pleson	2022-06-17
BARBARA LAKE AREA	596219	Single Cell Mining Claim	Alex Pleson	2022-06-17
BARBARA LAKE AREA	596218	Single Cell Mining Claim	Alex Pleson	2022-06-17
BARBARA LAKE AREA	596217	Single Cell Mining Claim	Alex Pleson	2022-06-17
BARBARA LAKE AREA	596216	Single Cell Mining Claim	Alex Pleson	2022-06-17
BARBARA LAKE AREA	596215	Single Cell Mining Claim	Alex Pleson	2022-06-17
BARBARA LAKE AREA	596214	Single Cell Mining Claim	Alex Pleson	2022-06-17
BARBARA LAKE AREA	547826	Single Cell Mining Claim	Alex Pleson	2021-04-8



SCHEDULE B

ROYALTY AGREEMENT

See attached.

ROYALTY AGREEMENT

THIS NET SMELTER RETURNS ROYALTY AGREEMENT dated as of the 5th day of August, 2020 .

BETWEEN:

1257590 B.C. LTD., a corporation incorporated pursuant to the laws of British Columbia (hereinafter referred to as the "**Owner**")

- and –

ALEX PLESON, an individual (hereinafter referred to as the "**Holder**")

THIS AGREEMENT WITNESSES THAT for good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged by each of the Parties, the Parties agree as follows:

1. **Definitions**

"Affiliate" has, with respect to the relationship between two or more companies, the meaning given to it in the *Business Corporations Act* (British Columbia) as such act may be amended from time to time and, with respect to the relationship between two or more Persons any of which are not bodies corporate, a Person shall be deemed to be an Affiliate of another Person if one of them is controlled by the other or if both are controlled by the same Person, and for this purpose, control means the right, directly or indirectly, to direct or cause the direction of the management of the affairs of a Person, whether by ownership of securities, by contract or otherwise.

"Allowable Deductions" means

- (a) all value added taxes or government charges based directly on or assessed against the existance, severance, production, removal, sale, processing, transportation, disposition, value or quantity of Minerals, including the following:
 - (i) direct sales tax;
 - (ii) use taxes;
 - (iii) gross receipts taxes;
 - (iv) severance taxes; and
 - (v) crown royalties,

but excluding any and all taxes based upon the net or gross income of the Owner and/or its Affiliates or other operator of the Property, the value of the Property or the privilege of doing business, and other taxes assessed on a similar basis;

- (b) all insurance costs;
- (c) all transportation costs (including related insurance, shipping, freight, stockpiling, storage, warehousing, handling, port, demurrage, delay and forwarding expenses, transaction taxes, customs and customs clearance, import and export duties and permit

costs) for transportation of Minerals from the Property to a Processor and from there to the point of sale;

- (d) all marketing and sales costs related to the sale of Minerals including commissions incurred in selling Minerals;
- (e) all costs, tolling charges, representation expenses, metal losses, umpire charges, expenses, penalties, fees, including all weighing, sampling and assaying costs and other expenses and charges of any nature whatsoever and including price participation charges (including in the form of retained metal) that are paid or incurred by the Owner for or in connection with smelting, refining, beneficiation processes or mineral treatment of the Minerals charged by any Processor against the Owner; and
- (f) if the Owner processes any Minerals using methods of processing on any of the Property (or at an offsite location(s) reasonably proximate to any of the Property) instead of, or prior to, transporting any of the Minerals to a Processor, in addition to the deductions set out in (d) above, the direct operating, maintenance and administrative cost of such onsite or near-site processing, including amortization of the capital (including sustaining capital) for such processing facilities over a straight line 20 year period, provided that any deductions for onsite processing may not exceed the charges and costs the Owner would have incurred if such processing was carried out at facilities that were not owned or controlled by the Owner and that are offering comparable services for comparable products;

"**Business Day**" means any day other than a Saturday, Sunday or day that is a statutory holiday in any of Vancouver, British Columbia or Toronto, Ontario.

"Governmental Authority" means any federal, provincial or local governmental, regulatory or administrative authority, agency or commission, or any court, tribunal or judicial or arbitral body, having jurisdiction.

"IFRS" means International Financial Reporting Standards.

"Minerals" means all marketable naturally occurring minerals or mineral bearing material in whatever form or state, including, without limitation, mineral extracted, removed, produced or otherwise recovered from the Property (but, for greater certainty, not including any salt, rock, sand, gravel or aggregate), whether in the form of ore, concentrates or any other beneficiated or derivative products thereof and including any mineral bearing materials or products derived from any processing or reprocessing of any tailings or other waste products originally derived from the Property but shall not include any material mined from the Property for use by Owner for roads, foundations, concrete or other construction or industrial uses relating to the Property.

"**Net Smelter Returns**" means the gross proceeds actually received by the Owner from the sale or other disposition of Minerals less Allowable Deductions.

"Party" means the Owner or the Holder and "Parties" means the Owner and the Holder, collectively.

"**Person**" means and includes individuals, corporations, limited partnerships, general partnerships, joint stock companies, limited liability corporations, joint ventures, associations, companies, trusts, banks, trust companies, pension funds, business trusts or other organizations, whether or not legal entities.

"Place of Delivery" means the place or account directed by the Holder in writing.

"**Processor**" means collectively any mill, smelter, refinery, mint or other processor of the Minerals, other than the Owner, which processes any Minerals at any stage before sale or other disposition by the Owner.

"Property" means the mineral titles described on Schedule "A" attached hereto.

"Royalty" means the payments to be made to the Holder described in Section 2 of this Agreement.

2. **Grant of Royalty**

- (a) Subject to the terms of this Agreement, the Owner, hereby grants, sells, assigns, transfers, conveys and agrees to pay to the Holder a royalty equal in aggregate to 2 percent (2%) of Net Smelter Returns, subject to the reduction contained in Section 2(b) (the "**Royalty**").
- (b) The Owner has the right, exercisable at any time by written notice to the Holder and by subsequent payment of the sum of \$500,000 (the "Buyout Payment") to the Holder, to reduce the percentage of Net Smelter Returns constituting the Royalty payable under this Agreement to 1% of the Net Smelter Returns from the Property. If such Buyout Payment is made, the Parties acknowledge that the Royalty will be reduced to 1% of the Net Smelter Returns from the Royalty will be reduced to 1% of the Net Smelter Returns from the Property.

3. Time and Manner of Royalty Payments

- (a) The Royalty shall be calculated and paid for each fiscal quarter of each calendar year during the term of this Agreement (a "quarter") (i.e., each succeeding three month period of a calendar year, the first quarter commencing on January 1st), commencing with the first quarter following issuance of a press release by the Owner that commercial production as prescribed in the Owner's feasibility study in respect of the Property has commenced. The Royalty paid in such first quarter shall be calculated based on the Net Smelter Returns received by the Owner prior to such quarter, including during any commissioning period. The Royalty for each guarter shall be paid to the Holder, or as the Holder may direct the Owner in writing, by the Owner by certified cheque, bank draft or wire transfer (in the sole and absolute discretion of the Holder) in Canadian dollars, on or before the day that is sixty (60) days after the last day of each quarter. Any adjustment to the determination of any Royalty payment shall be made on the next scheduled Royalty payment. All such Royalty and adjustment payments shall be delivered to the Holder at the Place of Delivery in such manner as specified in writing by the Holder.
- (b) At the time each Royalty payment is paid to the Holder, the Owner shall prepare and deliver to the Holder a statement setting out in reasonable detail the manner in which such Royalty payment was calculated, including: (i) the quantities of Minerals sold or otherwise disposed of by the Owner and the gross proceeds actually received by the Owner in connection therewith; (ii) the quantities of Minerals to which such Royalty payment is applicable; (iii) the calculation of the applicable Net Smelter Returns; and (iv) in the event of any commingling as contemplated in Section 5, a detailed summary of the determination by the Owner of the quantity of Minerals commingled in accordance with Section 5 and subject to the Royalty.

- (c) Notwithstanding the terms of any other provision in this Agreement, the Owner shall not be obligated to make any Royalty payment before the Owner has actually received the gross proceeds for the sale or other disposition of, Minerals.
- (d) The Holder may object in writing to any statement or Royalty payment amount within two (2) months of the receipt by the Holder of the relevant statement in respect of such payment. If it is determined by agreement of the Parties or by arbitration that any Royalty payment has not been properly paid in full as provided herein, the Owner shall prmprity thereafter pay the Holder the amount of the applciable shortfall in Royalty payment.
- (e) If it is determined by agreement of the Parties or by arbitration that any Royalty payment was overpaid, the Owner shall be entitled to offset such amount against the next Royalty payment.
- (f) All Royalty payments will be made subject to withholding or deduction in respect of the Royalty for, or on account of, any present or future taxes, duties, assessments or governmental charges or levies of whatever nature imposed or levied on such Royalty payment by or on behalf of any Governmental Authority having power and jurisdiction to tax and for which the Owner is obligated in law to withhold or deduct and remit to such Governmental Authority. The Owner shall set out in the statement referred to in Section 3(b) any amount so withheld.
- (g) All tailings or other waste products resulting from the mining, milling or other processing of ores derived from the Property from and after the date of this Agreement shall be the sole and exclusive property and responsibility of the Owner, but shall be subject to the Royalty and the terms of this Agreement, including the provisions in respect of commingling, if such tailings or other waste products are processed in the future resulting in the production of Minerals therefrom.

4. Term

This Agreement shall continue in perpetuity, it being the intent of the Parties that the Royalty shall constitute a covenant running with and binding upon the title to the Property and all accessions thereto and all successions thereof, whether created privately or through governmental action, and, subject to Subsection 13(a), binding upon the successors and assigns of the Owner and the successors in title to the Property. If any right, power or interest of either Party under this Agreement would violate the rule against perpetuities, then such right, power or interest shall terminate at the expiration of 20 years after the death of the last survivor of all the lineal descendants of Her Majesty, Queen Elizabeth II of England, living on the date of this Agreement.

5. **Commingling**

- (a) Subject to Subsection 5(b) below, the Owner shall be entitled to commingle Minerals from the Property and from any other property owned or leased by the Owner, during the stockpiling, milling (concentrating), smelting, refining, minting or further processing of Minerals produced from the Property, but, for greater certainty, not at any time prior to or during the mining phase of production.
- (b) Before any Minerals are commingled with ores or minerals from any other property, including stockpiling, the Minerals shall be measured and sampled in accordance with

standard mining and metallurgical practices. Representative samples of the Minerals shall be retained by the Owner and assays and appropriate analyses of these samples shall be made before commingling to determine mineral, moisture and other appropriate content of the Minerals. From this information, the Owner shall determine the quantity of the Minerals subject to the Royalty notwithstanding that the Minerals have been commingled with ores or minerals from other Property. Absent objection made by the Holder pursuant to Section 3(d), the Owner may dispose of the materials and data required to be produced and kept by this Section after a period of twelve (12) months from the date such materials and data are produced.

6. Hedging Transactions

All profits, losses and expenses resulting from the Owner engaging in any commodity futures trading, option trading, metals trading, metal loans, and any other hedging transactions or any combination thereof (collectively "**hedging transactions**") are specifically excluded from calculations of Royalty payments pursuant to this Agreement. All hedging transactions shall be for the Owner's sole account and shall not affect the calculation and payment to the Holder of the Royalty which shall be calculated and paid in accordance with Section 3 without regard for any hedging transactions.

7. Stockpiling

The Owner or operator shall be entitled to temporarily stockpile, store or place ores or mined rock containing Minerals produced from the Property in any locations owned, leased or otherwise controlled by the Owner or its Affiliates, or the Processor, on or off the Property, provided the same are appropriately identified and secured from loss, theft, tampering and contamination.

8. Books; Records; Inspections

- (a) The Owner shall keep complete and accurate books and records of all of its operations and activities with respect to the Property, including the mining of Minerals therefrom and the treatment, processing, refining and transportation of Minerals derived therefrom, prepared in accordance with IFRS consistently applied. Subject to complying with the confidentiality provisions of this Agreement, the Holder and/or its authorized representatives shall be entitled, at their sole expense and sole risk, upon delivery of three (3) Business Days' advance notice, and during the normal business hours of the Owner, to perform audits or other reviews and examinations of the Owner's books and records relevant to the calculation and payment of the Royalty pursuant to this Agreement no more than once per calendar year to confirm compliance with the terms of this Agreement, including without limitation, calculations of Net Smelter Returns. The Holder shall diligently complete any audit or other examination permitted hereunder. All expenses of any audit or other examination permitted hereunder shall be paid by the Holder.
- (b) In performing such audit the Holder and/or its agents shall have reasonable access to all sampling, assay, weighing, and production records, including all mining, stockpile, commingling and milling records of the Owner relating to the Property and any Minerals derived from the Property (and the Holder shall be allowed to make notes or a photocopy thereof), all of which such records shall be kept and retained by the Owner or operator of the Property in accordance with good mining industry practice for the period of retention set out in Section 5(b).

(c) The Holder and its representatives, at their sole expense and sole risk, shall be entitled upon delivery of three (3) Business Days' advance notice in writing to visit the Property and be granted reasonable access to the Owner's management on an annual basis, or more frequently if reasonably requested. While on the Property for any reason, the Holder and its representatives must comply with all of the Owner's policies and codes of conduct, including without limitation, policies with respect to health and safety.

9. **Confidentiality**

- (a) Neither Party shall, without the express written consent of the other (which consent shall not be unreasonably withheld), disclose any non-public information in respect of the terms of this Agreement or otherwise received by such Party under or in conjunction with this Agreement and, in the case of the Holder, concerning Minerals and operations on the Property or any other property owned or leased by the Owner, other than to its employees, agents and/or consultants for purposes related to the administration, or assignment by the Holder, of this Agreement and neither Party shall issue any press releases concerning the terms of this Agreement or, in the case of the Holder, in respect of the operations of the Owner, without the consent of the other Party after such Party having first reviewed the terms of such press release. Each Party agrees to reveal such information only to its employees, agents and/or consultants who need to know, who are informed of the confidential nature of the information and who agree to be bound by the terms of this Section 9. In addition no Party shall use any such information for its own use or benefit except for the purpose of enforcing its rights under this Agreement.
- (b) The Parties may disclose data or information obtained under or in conjunction with this Agreement and otherwise prohibited from disclosure by this Section 9 after providing the other Party with a copy of the proposed disclosure and if the other Party does not object, acting reasonably, to such disclosure by notice in writing to such Party within 48 hours after receipt of such copy:
 - (i) to any third person to whom such Party in good faith anticipates selling or assigning its interest hereunder;
 - (ii) to a prospective lender to such Party; or
 - (iii) to a prospective equity financier or investor of such Party;

provided that in each case the Party to whom disclosure is proposed shall first have been provided with and signed and delivered to the other Party a confidentiality agreement executed by such third party purchaser, lender, financier or investor which agreement shall include the confidentiality provisions of this Section 9.

(c) The Parties may disclose data or information obtained under this Agreement if required to do so for compliance with applicable laws, rules, regulations or orders of a governmental agency or stock exchange having jurisdiction over the Parties, provided that such Party shall disclose only such data or information as, in the opinion of its counsel, is required to be disclosed and provided further that it will provide the other Party with a copy of the proposed disclosure and the other Party shall be given the right to review and object to the data or information to be disclosed within 48 hours (or such shorter timeframe as may be required pursuant to applicable law) of its receipt of such copy prior to any release, and any such release will be subject to any reasonable objections or changes proposed by such other Party.

10. Conduct of Operations

- (a) All decisions concerning methods, the extent, times, procedures and techniques of any (i) exploration, development and mining related to the Property, (ii) leaching, milling, processing or extraction treatment and (iii) materials to be introduced on or to the Property or produced therefrom, shall be made by the Owner in its sole and absolute discretion, and all decisions concerning the sale or other disposition of Minerals from the Property, shall be made by the Owner.
- (b) The Owner shall not be responsible for or obliged to make any Royalty payments for Minerals or Mineral value lost in any mining or processing of the Minerals conducted in accordance with accepted mining practices.

11. No Implied Covenants

The Parties agree that there are no implied covenants or duties relating to or affecting any of their respective rights or obligations under this Agreement, and that the only covenants or duties which affect such rights and obligations shall be those expressly set forth and provided for in this Agreement.

12. Assignment by Holder

The Holder shall have the right, at any time and from time to time, to assign, transfer, convey, mortgage, pledge or charge all, but not less than all, of the Royalty and its interest in and to this Agreement, provided that: (i) prior to entering into a binding agreement in respect of the assignment or transfer of this Agreement, other than for the purpose of creating a security interest only, the Holder provides the Owner with thirty (30) days' prior notice in writing of the Holder's intention to assign or otherwise transfer or convey this Agreement; and (ii) any such assignee, transferee, mortgagee, pledgee or chargee enters into a written agreement with the Owner whereby it agrees to be bound, and to cause any assignee, transferee, mortgagee, pledgee or chargee from it to be bound, by the terms of this Agreement. The Holder shall notify the Owner in writing prior to the completion of any such assignment, transfer or conveyance, confirming the identity of such transferee, the appropriate Place of Delivery and the new address for notice to such transferee.

13. Transfer by Owner

- (a) The Owner shall be entitled to assign, sell, transfer, lease, mortgage, charge or otherwise encumber or dispose of the Property or the Minerals *in situ*, or any parts thereof, or the proceeds thereof, and its rights and obligations under this Agreement, provided the following conditions are satisfied, and upon such conditions being satisfied in respect of any such assignment, sale or transfer, the Owner shall be released from its obligations under this Agreement corresponding to its proportionate interest in the Property so sold, transferred or otherwise disposed of, other than by way of security interest only,
 - (i) any purchaser, transferee, lessee or assignee of the Property or this Agreement agrees in writing in favour of the Holder to be bound by the terms of this Agreement including, without limitation, this Section 13;

- (ii) any purchaser, transferee or assignee of this Agreement has simultaneously acquired the Owner's right, title and interest in and to the Property or the Minerals in situ, or part thereof; and
- (iii) any mortgagee, chargee, lessee, assignee or encumbrancer of the Property or this Agreement agrees in advance in writing in favour of the Holder to be bound by and subject to the terms of this Agreement in the event it takes possession of or forecloses on all or part of the Property and undertakes to obtain an agreement in writing in favour of the Holder from any subsequent purchaser, lessee, assignee or transferee of such mortgagee, chargeholder, lessee or encumbrancer that such subsequent purchaser, lessee, assignee or transferee will be bound by the terms of this Agreement including, without limitation, this Section 13.
- (b) For greater certainty, the Holder hereby acknowledges that it has no right, title or interest in and to the surface rights of the Property.

14. **Dispute Resolution**

- (a) In the event of a dispute in relation to this Agreement, including, without limitation, the existence, validity, performance, breach or termination thereof, or any matter arising therefrom, the Parties will attempt to resolve amicably any such dispute by referral to successively higher level of the Parties' respective management (as applicable). If there is no resolution of the dispute by this means within thirty (30) days, then such dispute can be resolved by pursuant to this Section 14.
- (b) Any Party to this Agreement may request that the dispute be resolved by binding arbitration, conducted in English through the British Columbia International Commercial Arbitration Centre, in Vancouver, British Columbia. The seat of arbitration shall be Vancouver, British Columbia.
- (c) To request arbitration, the moving Party shall give written Notice to the other Party (a "**Responding Party**"), which Notice shall toll the running of any applicable limitations of actions under any Applicable Law or under this Agreement. Such Notice shall specify the nature of the allegation and issues in dispute, the amount or value involved (if applicable) and the remedy requested.
- (d) Within 20 days of the receipt of the Notice (the "Response Period"), the Responding Party shall answer the demand in writing, specifying the allegations and issues that are disputed, and advising of any cross-claim or counterclaim that they intend to pursue in the arbitration. Any Party failing to respond in accordance with this Section shall not be entitled to further notice of, or participation in, the arbitral proceeding.
- (e) The arbitration shall be conducted a single arbitrator. The procedure for selecting the single arbitrator shall be as follows: each Party shall select an individual designee (the "**Designee**"), which such individual must be indicated as a panelist, arbitrator or mediator on the list of same maintained by the British Columbia International Commercial Arbitration Centre (the "**List**"). The two Designees so appointed shall, within 15 days of the expiry of the Response Period, mutually agree upon the arbitrator, who shall be a fellow panelist, arbitrator or mediator that is indicated as such on the List. If a Party fails to appoint a Designee within the Response Period, then the Party that did appoint a Designee within such period shall be entitled to select the arbitrator, which

such arbitrator may be the Designee, or another individual indicated as a panelist, arbitrator or mediator on the List. The arbitration shall be conducted in accordance with the British Columbia International Commercial Arbitration Centre's rules (the **"Rules**").

- (f) The arbitrator shall fix a time and place in Vancouver, British Columbia reasonably convenient for the Parties, after giving the Parties not less than seven (7) Business Days' notice, for the purpose of hearing the evidence and representations of the Parties and the arbitrator shall preside over the arbitration and determine all questions of procedure not provided for under the Rules or this Section. After hearing any evidence and representations that the Parties may submit, the arbitrator shall make a decision and reduce the same to writing and deliver one copy thereof to the Claimant and the Respondent. The arbitrator shall endeavor to make a decision within forty-five (45) days after its appointment, subject to any reasonable delay due to unavoidable circumstances. Any decision by the arbitrator shall follow and apply the laws applicable to this Agreement pursuant to Section 15(c). The expense of the arbitration, including travel costs, expert witness and legal fees and costs shall be paid as determined in the discretion of the arbitrator, having due regard for the outcome of the arbitration and the relationship of the result to the positions taken by the Parties. In the absence of fraud or manifest error, the decision of the arbitrator shall be final and binding upon each of the Parties and the Parties expressly exclude any and all rights to appeal, set aside or challenge any award by the arbitrator insofar as such exclusion can be validly made.
- (g) Judgment upon the award may be entered by any court having jurisdiction thereof or having jurisdiction over the relevant Party or its assets. Except where matters are expressed herein to be subject to arbitration, the provincial or federal courts sitting in British Columbia, Canada shall have exclusive jurisdiction to hear and determine all matters relating to this Agreement, including enforcement of the obligation to arbitrate.
- (h) Nothing in this Agreement shall prevent any Party from applying to the provincial or federal courts in British Columbia, Canada for interlocutory, injunctive, provisional, or interim measures, including but not limited to any claim for preliminary injunctive relief.
- (i) All papers, Notice or process pertaining to an arbitration hereunder may be served on a Party as provided in Section 15(g).
- (j) The Parties agree to treat as confidential, in accordance with the provisions of Section 9, the following: the existence of the arbitral proceedings; written Notices, pleadings and correspondence in relation to the arbitration; reports, summaries, witness statements and other documents prepared in respect of the arbitration; documents exchanged for purposes of the arbitration; the contents of any award or ruling made in respect of the arbitration. Notwithstanding the foregoing part of this Section 14(j), a Party may disclose such information in judicial proceedings to enforce, nullify, modify or correct an award or ruling and as permitted under Section 9.

15. General Provisions

(a) <u>Relationship of the Parties</u>

Nothing herein shall be construed to create, expressly or by implication, a joint venture, mining partnership, commercial partnership, or other partnership relationship between the Owner and the Holder, or in any way create a fiduciary duty of one Party to another Party.

(b) Assignment

Any assignment, transfer, conveyance, mortgage, pledge or charge or lease or purported assignment, transfer, conveyance, mortgage, pledge or charge or lease of any interest in the Property by the Owner, or in, to or arising under this Agreement by the Owner or the Holder, which does not comply with the terms of this Agreement shall be null and void and of no force or effect whatsoever.

(c) <u>Governing Law</u>

This Agreement shall be governed by and construed under the laws of the Province of British Columbia and the federal laws of Canada applicable therein, without regard to any conflict of laws or choice of laws principle that would permit or require the application of the laws of any other jurisdiction. Each of the Parties hereby irrevocably attorns and submits to the arbitral jurisdiction set forth in Section 14 and, with respect to any matters not determined by arbitration, to the exclusive jurisdiction of the courts of British Columbia, Canada respecting all matters relating to this Agreement and the rights and obligations of the Parties hereunder.

(d) <u>Time of Essence</u>

Time is of the essence in this Agreement.

(e) <u>Severability</u>

If any provision of this Agreement is wholly or partially invalid, this Agreement shall be interpreted as if the invalid provision had not been a part hereof so that the invalidity shall not affect the validity of the remainder of the Agreement which shall be construed as if the Agreement had been executed without the invalid portion. It is hereby declared to be the intention of the Parties that this Agreement would have been executed without reference to any portion which may, for any reason, hereafter be declared or held invalid.

(f) Accounting Principles

All calculations hereunder shall be made in accordance with IFRS as the same may be in effect from time to time.

(g) <u>Notices</u>

Any notice or writing required or permitted to be given under this Agreement or any communication otherwise made in respect of this Agreement (referred to in this Section as a "**Notice**") shall be sufficiently given if in writing and: (i) delivered personally, either to the individual designated below for such Party, or to an individual having apparent authority to accept deliveries on behalf of such individual at the address set out below for such Party; (ii) by registered mail to the address set out below for such Party; or (iii) transmitted by email, at or to the applicable email addresses set out below for such Party:

(1) In the case of Notice to the Holder, at:

Notice provision redacted.

(2) In the case of Notice to the Owner, at:

Notice provision redacted.

or at such other address as the Party to whom such Notice is to be given shall have last notified the Party giving the same, in the manner provided in this Section. Any Notice delivered to the Party to whom it is addressed as provided in this Section shall be deemed to have been given and received on the day it is so delivered at such address, provided that if such day is not a Business Day then the Notice shall be deemed to have been given and received on the Business Day next following such day. Any Notice mailed to a Party to whom it is addressed as provided in this Section shall be deemed to have been given and received on the third Business Day following the date of mailing, except in the event of disruption of the postal services, in which event, Notice will be deemed to have been given and received will be deemed only when actually received. Any Notice transmitted by facsimile or email shall be deemed given and received on the first Business Day after its transmission.

(h) <u>Schedules</u>

The schedules which are attached to this Agreement are incorporated into this Agreement by reference and are deemed to form part hereof.

[Remainder of page intentionally left blank.]

(i) <u>Execution in Counterparts and by Electronic Delivery</u>

This Agreement may be executed in any number of counterparts and by the different Parties hereto on separate counterparts, each of which when so executed and delivered shall be an original, but all such counterparts together shall constitute one and the same instrument. Such counterparts may be delivered by regular post, courier or electronic mail.

IN WITNESS WHEREOF the Parties have duly executed this Agreement as of the date first above written.

Signature redacted.

WITNESS

Signature redacted.

ALEX PLESON

1257590 B.C. LTD.

Per: Signature redacted.

Name: Title:

SCHEDULE A

DESCRIPTION OF THE PROPERTY

Township / Area	Tenure ID	Tenure Type	Registered and Beneficial Title Holder	Anniversary Date
BARBARA LAKE AREA	599765	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA,LAKE JEAN AREA	599764	Multi-cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599763	Single Cell Mining Claim	Alex Pleson	2022-07-19
LAKE JEAN AREA	599762	Single Cell Mining Claim	Alex Pleson	2022-07-19
LAKE JEAN AREA	599761	Single Cell Mining Claim	Alex Pleson	2022-07-19
LAKE JEAN AREA	599760	Single Cell Mining Claim	Alex Pleson	2022-07-19
LAKE JEAN AREA	599759	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599758	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA,LAKE JEAN AREA	599757	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA,LAKE JEAN AREA	599756	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA,LAKE JEAN AREA	599755	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599754	Single Cell Mining Claim	Alex Pleson	2022-07-19
BARBARA LAKE AREA	599753	Single Cell Mining Claim	Alex Pleson	2022-07-19